Author's response to reviews

Title: Development and evaluation of an instrument for the critical appraisal of randomized controlled trials of natural products.

Authors:

Tannis Jurgens (tannis.jurgens@dal.ca)  
Anne Marie Whelan (anne.marie.whelan@dal.ca)  
Melissa MacDonald (mmmacdon@dal.ca)  
Lindsay Lord (lordl@dal.ca)

Version: 5 Date: 21 April 2009

Author's response to reviews: see over
April 21, 2009

The Editor BioMed Central Complementary and Alternative Medicine

Re: MS: 1648499584241248
Development and evaluation of an instrument for the critical appraisal of randomized controlled trials of natural products.
Tannis Jurgens, Anne Marie Whelan, Melissa MacDonald and Lindsay Lord

In response to the recent email regarding the final proofing of our manuscript, we have read through the manuscript a final time and have found several typos that we wish to correct. We have uploaded the corrected manuscript. The following are a list of the corrections that were made to the manuscript:

1. p6. In section Phase 1. Establishment of a list....second line-added the word “from” between as and items.

2. p12 In Results section, Phase 1, Selection of items. 4th line. Item #14 was changed to item #15.

3. p14 In section “Compilation of items from...”, second paragraph, second line “items” was changed to “item”

4. p14 In section “Compilation of items from...”, second paragraph, line 6 “#7. The study was blinded and item 24 The success of blinding was evaluated” was changed to read “item #7 (The study was blinded) and item #24 (The success of blinding was evaluated).”

5. p19 Second paragraph (before section titled: Evaluation of instrument) last sentence space added between “Delphi consensus and ref 28”

6. p20 First line on page. A comma was added after “pharmacy students”.

7. p20 second paragraph, last sentence. The word “improve” was changed to “reduced”.

Dr. Tannis Jurgens
College of Pharmacy, Dalhousie University
5968 College Street
Halifax, Nova Scotia
Canada, B3H 3J5
Ph: (902) 494-3796 Fax: (902) 494-1396
Email: tannis.jurgens@dal.ca
We are confident that the manuscript is now ready for publication.

Sincerely,

Tannis Jurgens  Ph D
Associate Professor, Medicinal Chemistry and Natural Health Products